METHODOLOGY

Open Access

Reporting involvement activities with children and young people in paediatric research: a framework analysis

Jennifer Preston^{1*}, Giovanni Biglino², Victoria Harbottle^{3,4}, Emma Dalrymple⁵, Helen Stalford⁶ and Michael W. Beresford^{1,7}

Abstract

Background The active involvement of patients and the public in the design and delivery of health research has been increasingly encouraged, if not enforced. Knowledge of how this is realised in practice, especially where children and young people (CYP) are concerned, is limited, partly due to the low level of reporting of patient and public involvement (PPI) in general. The aim of this work was to assess how researchers funded by the National Institute for Health and Care Research (NIHR) report the involvement of CYP in the design and conduct of child health research to better understand the opportunities offered to CYP, and the realities of involvement in practice.

Methods A participation matrix, analysis framework and accompanying tools were adapted from existing frameworks, including a child-rights informed framework, the Guidance for Reporting Involvement of Patients and the Public Checklist Short Form (GRIPP2SF), and NIHR reporting expectations. Child-focused research reports were identified from the NIHR Journals Library, including any interventional or observational study involving CYP aged 0–<24 years. In two co-design workshops with healthcare professionals and CYP, we tested and refined the participation matrix, analysis framework and accompanying tools.

Results Only thirty-two NIHR reports out of 169 (19%) were identified as relevant and included reporting of PPI with CYP. We identified significant variability in the way PPI with CYP was reported. Only 4/32 (12%) reports fully met NIHR (and GRIPP2SF) reporting criteria. Only 3/32 (9%) reports formally evaluated or self-reflected on PPI activities with CYP, whilst 15/32 (47%) provided minimal information about CYP involvement. The most common approach to involving CYP (23/32, 72%) was through the medium of existing groups or networks.

Conclusion Despite the NIHR's commitment to increase the quality, transparency, and consistency of reporting PPI, the reporting of involvement with CYP remains sub-optimal. Neglecting to report key details of involvement methods and impacts deprives the research community of knowledge to advance the field of delivering 'meaningful' PPI with CYP. Practical guidance on how researchers can report the processes and outputs of CYP involvement more rigorously may help child health researchers to involve them more meaningfully. This research offers practical tools informed by CYP to aid the reporting process.

*Correspondence: Jennifer Preston jennifer.preston@liverpool.ac.uk Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain English Summary

Background Children and young people can (and should) be involved in the design of child-health research. How this works in practice is limited due to the low-level reporting of involvement activities.

Aim We wanted to understand how researchers funded by the National Institute for Health and Care Research (NIHR) are reporting involvement activities with children and young people in child-health research.

What we did and how we did it: A team of healthcare researchers (working in child-health research), a parent, and children and young people adapted existing tools to assess how the involvement of children and young people is reported in child-health research. We looked at reports in an online library to see how researchers are meeting the reporting guidelines set by the NIHR.

Findings Despite the reporting guidelines provided by the NIHR, only four out of thirty-two reports followed this. It is still unclear who is involved, why, and what outcomes and impact involvement has on the research process, on children and young people and/or on researchers. This is a missed opportunity to learn from previous experiences and to build on good practice. Further work is needed to understand why and how the NIHR, and other funders of health research collect such information, and what their plans are for sharing the findings with the wider research community.

Background

Active involvement of patients and the public in the design and delivery of health research, rather than as 'subjects' of research has been encouraged (if not required) for many years through policy [1, 2], regulations [3, 4], funders, [5, 6] and best practice guidance [7-9].

Patient and public involvement (PPI) encompasses initiatives to include patients, family members, carers and members of the public in developing and improving health services and medicines [10]. There are many definitions used to describe PPI interactions, which are often used interchangeably. These include, '*participation*' (when individuals take part in the actual research, [11] '*engagement*' (when research information is shared with the public, e.g. at research open days or on social media [11–13]), and '*involvement*' most commonly defined as '*research being carried out 'with' or 'by' members of the public rather than 'to', 'about' or 'for' them*' (indicating a more active collaboration between patients, public and researchers) [11].

In recent years, there has been a growing call to include the voices of CYP in societal decision-making (including in healthcare and health research) in order to ensure that policies and programmes are more responsive and relevant to the concerns and needs of CYP [14, 15]. Meaningful involvement is considered to be a fundamental human right as articulated in the United Nations Convention on the Rights of the Child (UNCRC) [16]. The UNCRC is binding on the government and on public authorities at all levels in the UK and provides a strong ethical framework for planning and delivering meaningful involvement of CYP.

However, defining how meaningful PPI is realised in practice, especially where child and young patients and the public are concerned is limited [17-19], partly due to the low level of reporting PPI in general [20]. Recent attempts to synthesise what is known from the literature about PPI with CYP concluded that improvements need to be made to the evaluation and reporting of PPI with CYP, in order for researchers and funders to better understand the different levels and roles CYP have, and hence what works best for them, in different settings, and what impact their involvement has on the actual research itself and on those who get involved [21-23]. Thus the resulting evidence base remains disjointed, and shared learning from previous experiences is lost, which potentially is a waste of resource that could otherwise be put to informing 'meaningful' involvement of CYP [21].

To address the issue of poor reporting of PPI, in April 2018 the National Institute for Health and Care Research (NIHR) advised authors of research it funds to refer to the Guidance for Reporting Involvement of Patients and the Public Checklist (GRIPP2) [24] to enhance the quality, transparency, and consistency of reporting PPI activities. The NIHR is one of the UK's leading funding bodies that has provided vital strategic and infrastructure support to embed PPI across publicly funded research, creating an environment that views PPI as a crucial component of the research process [25]. It is one of the first health research funders to publish comprehensive accounts of its commissioned research within its own publicly and permanently available journals. The NIHR Journals Library¹

¹ https://www.journalslibrary.nihr.ac.uk/about-us/

comprises a suite of five open access peer-reviewed journals reporting results from several of its NIHR Programmes, which address a range of health research priorities. Reports published in the NIHR Journals Library provide a full account of the research project, including methods and a full description of the results. Further, in keeping with its role of providing a comprehensive archive of funded research, all reports in the library should include an explanation of how patients and the public have been involved in the study. Authors are encouraged to report faithfully on PPI activities undertaken, even if only to acknowledge the absence of it within the study. The NIHR reporting expectations require authors of reports to describe the following: if there was no PPI in the study to state this in the report, setting out why this was not thought appropriate or was not feasible; what form the PPI took and at what stages it occurred during the study; what impact PPI had during the study and how it was useful; if there was little/no impact of PPI during the study, to state it; and the way(s) PPI supported or will support dissemination of the results.

Furthermore, the GRIPP2 Checklist was introduced as a guide for authors. The Checklist was developed using the EQUATOR method [26] for developing reporting guidelines. The Checklist consists of two forms; a short form (SF) version referred to as GRIPP2-SF used primarily for studies where PPI is a secondary focus, and includes five items on aims, methods, results, outcomes, and critical perspective. The long form (LF) version referred to as GRIPP2-LF is aimed at studies where the main focus is PPI, and includes thirty-four items on aims, definitions, concepts and theory, methods, stages and nature of involvement, context, capture or measurement of impact, outcomes, economic assessment, and reflections.

As of the 1st of April 2022, it is mandatory for all reports to include a separate section on PPI as a subheading in the discussion section covering details of the PPI approach (or justification for no PPI), and what impact this had on all aspects of the study.

As a result of the reporting guideline changes within NIHR, we decided to explore how CYP's involvement in the design and conduct of paediatric research (in any type of intervention, comparison, or outcome) is reported within the NIHR Journals Library. In this report we use the term 'involvement activities' to define the inclusion of CYP taking part in research advisory roles (as part of research advisory, focus or steering groups, etc.), advising on various or all aspects of study design and conduct.

Aim

The aim was to examine in detail reports that are completed by researchers about involvement activities with CYP. Only sections of reports that described the involvement of CYP in these types of activities, not adult involvement (parents and other stakeholders) were included in the analysis. We wanted to explore:

- 1. Who is involved and how?
- 2. What are the reported opportunities offered to CYP, including levels of involvement in the different phases of the research process?
- 3. What are the reported outcomes and impacts of PPI with CYP?
- 4. Do CYP support the dissemination of research findings and how?
- 5. What are the reported challenges and facilitators to involvement?

Ultimately, we analysed reports published in the NIHR Journals Library in 2018 up to 28th February 2022 to build a picture of how NIHR-funded researchers are reporting PPI with CYP. We acknowledge that documentary evidence is not a proxy for actual PPI practice. It is not our intention to offer judgements about practice or individuals responsible for writing the reports.

Methods

Step 1: Participation matrix and assessment tool.

To assess the reported opportunities offered to CYP including the levels of involvement across the various phases of the research process, we adapted an existing Participation Matrix (hereon referred to as a matrix) originally developed by Lansdown [27] (See Fig. 1) to promote the participation rights of CYP. The matrix has predominantly been used in social science research/ policy and practice. It was originally created to acknowledge that CYP can participate in activities, processes, and decision-making in three different ways: consultation, collaboration or child led. It was designed not to be hierarchical in nature but to be used to reflect different degrees of empowerment and influence that are all legitimate and appropriate depending on the context of the research [28, 29]. To accompany the matrix a rating and colour coding tool was developed by the authors to aid the assessment of the level of involvement described in each report (See Table 1). The rating and colour coding tool was tested on a selection of reports (n=7) until agreement was reached between the authors.

How much decision-making power or control over the research do children have?

| | | Children are not involved | Consulted | Collaborate with adult researchers | Child initiated or directed |
|-------------------------------|--|------------------------------|--|---|--|
| P | Prioritising or identifying the right research questions | | Children asked about problems that concern them. | Children and adults jointly prioritise or identify research to address problems that concern them. | Children choose their own research. |
| nases of the research process | Designing the research | | Children consulted about the design of the research. | Children and adults deliberate and jointly decide on the design of the research. | Children decide how the research is designed. |
| | Managing the research | | Children consulted in the management and conduct of the study. | Children and adults jointly steer the project throughout the research process. | Children direct with support from adults how the research is conducted ¹ |
| | Analysis and interpretation | | Adults show preliminary findings to children and ask for feedback. | Children and adults work together to analyse data and determine conclusions. | Children analyse data and draw their own conclusions. |
| | Dissemination of study findings | | Adults consult children on how to disseminate findings. | Children and adults collaborate on dissemination and awareness- raising activities. | Children undertake activities to disseminate their findings. |
| | Implementation | | Children consulted about how findings from the research can be implemented. | Children and adults work together on plans for implementing findings. | Children develop and implement an action plan to ensure research is taken into practice. |
| ł | Monitoring & Evaluation | | Children consulted about what they have learned from the process of involvement and impact on the research. | Children and adults work together to monitor and evaluate lessons learned from the process of involvement and impact on the research. | Children develop their own tools to monitor and evaluate lessons learned from the process of involvement and impact on the research. |

¹ Within the boundaries of ethics and law.

Fig. 1 Participation Matrix

 Table 1
 Rating and colour coding tool

| Rating | Colour | Definition of analysis |
|---------------|--------|---|
| Unmet | Red | No evidence of involvement |
| Partially met | Amber | Some evidence to suggest involvement |
| Fully met | Green | Clear Enough evidence of involvement Clear explanation and detail |

Step 2: Analysis framework and rating tool.

We then created and iteratively tested an additional analysis framework (See Table 2) and rating tool (See Table 3) to assess the remaining NIHR reporting expectations. The Framework Method [30] which is commonly used for content analysis was used to organise data into a framework where rows represent cases and columns support codes. By using this method, we could compare vertically between NIHR expected reporting requirements (and GRIPP2 SF) and horizontally across each report.

Search

A search took place from October 2021 to February 2022, using the advanced search option within the NIHR Journals Library. Individual search terms ("children", "child", "young people", "adolescents", "infants") were used to identify published paediatric focused reports, for any research type (primary research, evidence synthesis and methodology research), and any health category. Considering the implementation of the GRIPP2 guidance, the search parameters were reports published in 2018 up to 28th February 2022.

Selection criteria

Inclusion criteria

Reports were included in the review if they met the inclusion criteria outlined in Table 4. Any interventional or observational study was included if the study population included CYP between the ages of 0-<24 years, and the study involved this age group in any PPI activities. The age range was based on the World Health Organisation [31] definition of a 'child' as

Table 2 Analysis Framework

| ltem | Questions to support analysis |
|------------------------------------|--|
| Aim* | (1) Did the authors describe the aim(s) of PPI in the study? |
| Methods | (2a) Is there a description of the methods used for involving CYP at different phases of the research process? This would include a description of: Level of involvement (e.g., consultation, collaboration, child-led) PPI population (e.g., numbers involved, age group, other demographics, and medical condition) Model of involvement (e.g., development of a Project Advisory Group, focus group, tapped into existing Young Person's Advisory Group, etc.) (2b) Is there a description of the level of involvement within the various phases of the research process? This would include a description of: What phase of the research process are CYP involved? Is this involvement at just one phase of the research process or multiple levels? |
| Study results (outcomes)* | (3) Is there a description of the outcomes of PPI with CYP? (both positive and negative) |
| Impact | (4) Is there a description of the impact of PPI with CYP? (both positive and negative). This could include any description of the following impact: Impact on the study Impact on CYP Impact on researchers Impact on policy |
| Dissemination | (5) Is there a description of how CYP supported (or will support) the dissemination of research findings/results? |
| Reflections/critical perspectives* | (6) Is there a description of the conclusions and lessons learned from PPI with CYP? |

*Only required when reporting against the GRIPP2SF

Table 3 Rating Tool

| Rating | Code | Definition of analysis |
|---------------|------|--|
| Unmet | U | No evidence to address analysis questions |
| Partially met | Ρ | Unclear Some evidence to address analysis questions Poor explanation and detail |
| Fully met | F | Clear Enough evidence to address analysis questions Clear explanation and detail |

Table 4 Inclusion and exclusion criteria

Inclusion criteria

Study population included children and young people between the age range of 0-<24 years PPI activities occurred with children and young people aged 0-< 24 years Reports published in 2018 up to 28th February 2022 Any interventional or observational study Exclusion criteria Not relevant (study population didn't include 0-< 24 years) Reports waiting to be published Reports that do not mention PPI Reports that do not include children and young people in PPI activities Unable to separate CYP population (as PPI advisors) from other populations (for example, parents or carers) Insufficient information on PPI Describes 'engagement' not involvement Hard to distinguish between the actual research methods and PPI

a person under the age of 18 years and a 'young person' as under the age of 24 years of age (10-24 years). Reports published in 2018 up to 28th February 2022.

Exclusion criteria

Exclusion criteria are summarised in Table 4. Reports were excluded if they didn't include the study population, were still waiting to be published, did not mention PPI at all; did not include CYP in PPI reported activities, were unable to separate the CYP population (as PPI advisors) from other populations, described engagement rather than involvement activities, were hard to distinguish between qualitative research being undertaken and PPI activities specifically, and if there was insufficient information on PPI to assess according to the matrix.

Selection procedure

Reports were independently screened by the lead author (JP) in two stages using title, abstract, and keywords initially, and then the full text manuscript. After removing duplicates, reports that were clearly irrelevant were excluded in the first stage of title and abstract review, and if the eligibility of the report was not clear, the full text was then reviewed in the second stage. The second stage included extracting any data that described PPI with CYP in the report, including footnotes, acknowledgements or links to peer reviewed journals that described PPI activities.

Charting the data

A Microsoft Excel ® version 2021 was developed to assist in extracting and analysing the information on CYP involvement within the published reports. Core data was extracted to characterise the cohort and to describe any CYP activity throughout the duration of the research study. Study characteristics included (a) funding category; (b) research type; (c) health category; (d) health research activity code; (e) age range of study cohort; (f) start date of study; (g) end date; (h) date study was published. Extracted data relating to PPI activities with CYP included (a) reported aims of PPI; (b) reported level of involvement within the various phases of the research process, and roles given (how CYP were involved, and what CYP did); (c) reported outcomes and impact; (d) reported dissemination plans; and (e) any reflections on lessons learnt (both positive and negative).

Creating the tools to assess the comprehensiveness of reported involvement

(i) Assessing the level and opportunities offered to CYP

Once the data were extracted, methods to judge the comprehensiveness of the description of the reported involvement activities were required. The matrix (Fig. 1) and accompanying rating tool (Table 2) was used to assess the reported level and opportunities offered to CYP across the different phases of the research process, as advised by the NIHR [32] (excluding the commissioning phase of research). Using the simplified typology of involvement identified by Lansdown (children are consulted, children collaborate with adult researchers, or child-led) we also added a fourth column that captured where CYP are not involved.

(ii) Co-design phase

A small sub-group of healthcare professionals and a parent/PPI practitioner (co-authors GB, VH, ED) from the NIHR Paediatric Methodology Incubator (Patient and Public Involvement Working Group) was formed and led by the lead author (JP) to adapt and test the matrix, analysis framework, and accompanying assessment tools on the same seven reports used in step 1 of the methods. The subgroup met monthly for the duration of the project. Two additional workshops took place (one with five healthcare professionals from the wider team within the NIHR Paediatric Methodology Incubator, and one workshop with seven young people from the GenerationR Liverpool Young Person's Advisory Group (YPAG)). Although the focus of both workshops was the same, each workshop adopted a slightly different approach. The healthcare professionals' workshop (lasting 1 h 30 min) took place online. Participants were split into two groups, one facilitated by JP and the other facilitated by GB and VH. Each group received two PPI reports (of the seven selected reports used within the sub-group) in advance of the meeting accompanied by instructions on how to use the tools. We systematically captured and amended the wording of the matrix, analysis framework and accompanying tools considering feedback received during the workshop.

JP then facilitated a four-hour young person's workshop which took place face-to-face. The seven young people included five boys (aged 13, 14, 16 (\times 2), and 17 years old), and two girls (aged 17 and 18 years old). Each had been a member of the YPAG between 2 and 8 years. The workshop was designed to be interactive and was broken into two sessions and four activities, including a final discussion session (See the full YPAG Agenda in Additional file 1). The young people felt the matrix, analysis frameworks, and assessment tools were helpful to both reflect on how CYP are involved in research processes, and as helpful guides for research teams to adequately report PPI activities with CYP. The group did not suggest any further changes to the tools. Full details of these workshops will be presented in a separate paper.

Once the matrix, analysis framework and assessment tools were agreed the lead author then assigned these criteria independently for all reports that met the inclusion criteria. Further discussion was initiated with the sub-group if the lead author struggled to interpret the involvement description, for example if researchers were describing the qualitative aspect of the research, as opposed to describing how the involvement activities informed and supported the design of the qualitative methods.

NVivo 12 software was used to store, order and code data and to select supporting quotes. A rating was then assigned to each report and corresponding domain coding cell.

Findings

Search results

The initial search yielded 545 records, of which 351 were duplicates (see Fig. 2). Twenty-five reports were removed instantly as the study population was not relevant (n=3), and n=22 reports were waiting to be published. The search identified n=11 reports published before April 2018; once assessed for eligibility, only four of these reports were included in the final data synthesis and



Fig. 2 Schematic representation of report selection procedure

analysis. In total 169 full reports were retrieved, of which 137 were excluded for the reasons identified in Fig. 2. In total, thirty-two reports were included in the final review.

Characteristics of included reports

Details of the thirty-two reports are provided in Additional file 2. A summary is described below:

Twenty-seven of the thirty-two reports were primary research studies and five evidence synthesis studies. The primary research studies focused on the following health categories: cancer (n=2); congenital disorders (n=1); generic health relevance (n=5); infection (n=1); inflammatory and immune system (n=1); mental health (n=8); metabolic and endocrine (n=2); musculoskeletal (n=1); neurological (n=1); oral and gastrointestinal (n=2); reproductive health and childbirth (n=3). The five evidence synthesis studies focused on mental health (n=2) and generic health relevance (n=3).

How is PPI undertaken with children and young people?

The most common approach to involving CYP was through the medium of existing groups or networks, with 72% (23/32) choosing this approach. Of these 25, 60% (15/25) chose to access existing Young Person's

Advisory Groups (YPAGs) who have a remit to support the design and conduct of child health research [33-47], whilst 28% (8/25) reported some form of dialogue with existing youth forums such as, school advisory councils or care groups [36, 38, 39, 48-51], including one study accessed individuals through a PPI network within a Sexual and Reproductive Health Service (no description was provided within the report of who was involved in the network) [52]. 25% (8/32) established their own formal advisory group [53-60]. Of the remaining studies two set up focus groups or meetings with CYP [55, 60], two indicated they held one-off individual consultations with CYP [56, 61], two involved CYP in Study Steering Committees [53, 55], one reported involving young students in research training events held within a university [58], one reported involving CYP as mystery shoppers [52], and one report wasn't clear on the chosen model [62].

What opportunities are offered to children and young people?

Although not all reports explicitly described the chosen level of CYP involvement or at the precise phase of research, using the adapted matrix (Fig. 1), and accompanying rating tool allowed us to assess the opportunities offered to CYP within each research project. This was also aided by a description of tasks associated to

| | | Not involved | Consulted | | Collaborate with adults | Initiate or direct the research |
|--------|--|--------------|-----------|---|----------------------------|---------------------------------------|
| Phases | Prioritising or identifying the right research questions | 31 | 1 | | 0 | 0 |
| of th | Designing the research | 1 | 20 | 2 | 9 | 0 |
| e rese | Research conduct and operations | 30 | 0 | | 2 | 0 |
| arch | Analysis and interpretation | 25 | 5 | 1 | 1 | 0 |
| proces | Dissemination of study findings | 24 | 5 | 1 | 2 | 0 |
| SS | Implementation | 29 | : | 2 | 1 | 0 |
| | Monitoring & Evaluation | 30 | | 2 | 0 | 0 |

How much decision-making power or control over the research do children have?

Fig. 3 Final assessment of reported levels of CYP involvement

each phase of the research (See Additional file 3). Figure 3 summarises the final assessment of the reported levels of CYP involvement across the different phases of research highlighted in the reports.

At the phase of prioritising or *identifying the research question phase*: only one report [56] partially met the criteria for consulting CYP during this phase, and this was further evidenced in the published journal article separate from the main report [63]. The remaining thirty-one reports provided no evidence of involvement at this phase.

At the phase of *designing the research phase*: 97% of the reports (31/32) described varying levels of involvement in the design stage of the research. 71% of these reports (22/31) described involvement at the consultation level, with only 9% (2/22) fully meeting the criteria [38, 40], the remaining 91% (20/22) partially meeting the criteria [33, 34, 36, 37, 42–52, 54, 60–62, 64]. Instead, 29% fully met the criteria (9/31) describing a collaborative approach [39, 41, 53, 55–59, 65].

At the phase of *managing the research phase*: only 6% of the reports (2/32) highlighted a collaborative approach to involving CYP in the management of research, one via a young person being involved in the Study Steering Committee, [55] and one indicating direct linkage between a Children's and Young Person's Advisory Group to the Study Steering Committee [53].

At the phase of analysis and interpretation phase: 22% of the reports (7/32) indicated varying degrees of involvement in the analysis and interpretation phases of the research. Only one report fully met the criteria describing a collaborative approach to involving CYP

in this phase [65]. The other (6/7) described consulting with CYP, five of which partially met the criteria [34–36, 46, 55], and one fully met the criteria [53].

At the phase of *dissemination of study findings phase*: 25% of the reports (8/32) described varying degrees of CYP involvement in the dissemination phase of the research. 62% of these reports (5/8) partially met the criteria that described a consultation approach to involving CYP at this phase [35, 36, 52, 57, 60]; one report fully met the criteria in the consultation category [56]; two reports fully met the criteria describing a collaborative approach [53, 65].

At the phase of *implementation phase*: 9% of the reports (3/32) indicated involvement in the implementation phase of the research. 67% of these reports (2/3) partially met the criteria at a consultation level [58, 65], and one fully met the criteria at a collaborative level [53].

At the phase of monitoring and evaluation phase: only 6% of the reports (2/32) fully met the criteria for involvement at this phase of the research, both at a consultation level [53, 65].

Who is involved?

A full description of the demographics of CYP (gender, age, ethnicity, or health conditions versus healthy CYP) was weak for most reports. 84% (27/32) of the reports mentioned the age ranges of the CYP involved, the remaining five were less clear on the ages but indicated that primary school aged children were involved (n=2), or secondary school pupils or teenagers were involved (n=3). 96% (26/27) of the reports that mentioned age

ranges implied the involvement of CYP between the ages of 7–25 years old. One report involved a six-year-old in a one-off exercise. It is not clear from the reports of the breakdown of ages for each activity how many younger children are involved compared to adolescents or young adults.

Only 9% of the reports (3/32) mentioned the gender breakdown of involved CYP [35, 39, 56]. Only 6% (2/32) of the reports mentioned the ethnicity of CYP [39, 56].

18% of the reports (6/32) referred to CYP having specific conditions or experiences including: mental health conditions (3/6) [50, 55, 65] such as, social anxiety disorder and experience of eating disorders; experience of having dental treatment (1/6) [62]; cancer (1/6) [56], and experience of appendicitis (1/6) [57]. 18% of the reports (6/32) addressed more general health experiences, such as the experience of living with long term conditions (2/6) [56, 65]; accessing health-care services who had experience of living with physical and/or developmental conditions relevant to the study (2/6) [41, 53], interested in medical research (1/6) [31], and one with experience of intensive care (1/6) [61].

Frequency of involvement and types of activities

The frequency of involvement (see Fig. 4) ranged from a one-off consultation up to nine consultations. One report indicated they met regularly with CYP (monthly)



over the course of 5 years [53], and one report highlighted they held regular meetings (unknown amount) over the course of 10 years [56]. Both reports were large programme grants highlighting involvement in all workpackages and throughout the duration of the programme.

Quality assessment of reports

The length of text to describe PPI with all stakeholders (parents, charities, adults, CYP, etc.) varied from one short paragraph to full chapters within the main report or appendices. Three reports had also published about their involvement with CYP in journal articles, separate from the main published report. 12% (4/32) reports provided detailed information but blurred lines between PPI activities with actual study methods, making it difficult to determine who was involved and how.

Only 12% of the reports fully (4/32) met all the criteria for reporting PPI against the NIHR reporting expectations (See Table 5), and all four reports also met all the GRIPP2SF criteria.

Aim of research

Although the aim of PPI was not a NIHR mandatory field to complete, 22% of the reports (7/32) fully met this criterion [40, 41, 44, 53, 56, 61, 65] and one report partially met this criterion [55].

Methods

(a) Level and model

Only 25% (8/32) fully met the criteria for reporting the level and model of involvement [36, 41, 53, 56–59, 65]. The remaining reports partially met this criterion.

(b) Phase of research

40% (13/32) fully met the criteria for reporting involvement at different phases of the research process [34, 36, 39–41, 53, 55–57, 59, 60, 64, 65]. The remaining reports partially met this criterion.

Study results (outcomes)

41% of the reports (13/32) fully met the criteria describing the outcomes of PPI with CYP on study results [34, 38–41, 53, 55–57, 59, 60, 64, 65], 25% (8/32) partially met the criteria [45, 47, 49, 52, 54, 58, 61, 62], and the remaining 34% (11/32) did not meet the criteria. The most commonly reported outcomes focused on the positive impact on study design. These included: confirming the study importance, and study interventions and suitability of

| Report ID | Author | Aim* | (A) Methods (level & model) | (B) Methods (phase of research) | Outcomes* | Impact | Dissemination | Reflections* |
|-----------|---------------------------|------|--------------------------------|---------------------------------------|-----------|--------|---------------|--------------|
| NIHRJL01 | Moore et al. [65] | F | F | F | F | F | F | F |
| NIHRJL02 | Giles et al. [60] | U | Р | F | F | F | Р | Р |
| NIHRJL03 | Colver et al. [53] | F | F | F | F | F | F | F |
| NIHRJL04 | Blair et al. [33] | U | Р | Ρ | U | U | U | U |
| NIHRJL05 | Goodwin et al. [34] | U | Р | Ρ | F | F | U | U |
| NIHRJL06 | King et al. [54] | U | Р | Р | Р | Р | U | U |
| NIHRJL07 | Tancred et al. [35] | U | Р | Ρ | U | U | U | U |
| NIHRJL08 | Ford et al. [48] | U | Р | Р | U | U | U | U |
| NIHRJL09 | Bonell et al. [36] | U | Р | F | U | U | U | U |
| NIHRJL10 | Ramanan et al. [37] | U | Р | Ρ | U | U | U | U |
| NIHRJL11 | Creswell et al. [55] | Р | Р | F | F | F | U | F |
| NIHRJL12 | Mitchell et al. [64] | U | Р | Ρ | F | F | U | U |
| NIHRJL13 | Ponsford et al. [38] | U | Р | F | F | F | U | U |
| NIHRJL14 | Alderson et al. [39] | U | F | F | F | F | U | U |
| NIHRJL15 | Bray et al. [40] | F | Р | F | F | F | U | Р |
| NIHRJL16 | Janssens et al. [49] | U | Р | Ρ | Ρ | Ρ | U | U |
| NIHRJL17 | Maguire et al. [62] | U | Р | Ρ | Ρ | U | U | U |
| NIHRJL18 | Robling et al. [41] | F | F | F | F | F | F | F |
| NIHRJL19 | Meiksin et al. [42] | U | Р | Ρ | Ρ | U | U | U |
| NIHRJL20 | Mallucci et al. [43] | U | Р | Ρ | U | U | U | U |
| NIHRJL21 | Caldwell et al. [44] | F | Р | Р | Р | U | U | U |
| NIHRJL22 | Langton Hewer et al. [45] | U | Р | Ρ | U | U | U | U |
| NIHRJL23 | Taylor et al. [56] | F | F | F | F | F | F | F |
| NIHRJL24 | Hall et al. [57] | U | F | F | F | F | Р | F |
| NIHRJL25 | Cameron et al. [52] | U | Р | Р | Р | U | U | U |
| NIHRJL26 | Byford et al. [50] | U | Р | Р | U | U | U | U |
| NIHRJL27 | Tume et al. [61] | F | Р | Р | Р | U | U | U |
| NIHRJL28 | Kidger et al. [46] | U | Р | Р | U | U | U | U |
| NIHRJL29 | Adab et al. [51] | U | Р | Р | U | U | U | U |
| NIHRJL30 | Bee et al. [58] | U | F | F | Р | Р | U | U |
| NIHRJL31 | Cottrell et al. [47] | U | Ρ | Р | U | U | U | U |
| NIHRJL32 | Griffiths et al. [59] | U | F | F | F | F | U | U |

Table 5 Rating assigned to each report against NIHR reporting expectations

*Only required by GRIPP2SF

the research questions [56]; developing outcome measures important to CYP [56–58, 65], and core outcome set development [57]; contributing to the intervention development [64]; input into study documentation, including contributing to the design of interview topic guides and manuals aimed at CYP [39, 55, 59, 60], and survey development [56]; recruitment and retention materials, such as study information sheets or consent documentation, letters aimed at patients, videos, and advised on how to increase interest (of either PPI or research participation

Table 6 Example table used to describe end-user involvement

| Activity | Date | Who? | End-user perspective represented | Impact |
|----------------------|------|------|--|--------|
| Planning stage | | | | |
| Review methods stage | | | | |
| Consultation stage | | | | |
| Dissemination stage | | | | |
| | | | | |

| Table 7 | Example | table u | used to | hiahlia | aht findiı | nas from | PPI activities |
|---------|---------|---------|---------|---------|------------|----------|----------------|
| | | | | | / | | |

| Component/resource | Issues identified | Action taken to incorporate | |
|--------------------|-------------------|-----------------------------|--|
| | | | |

in the study) [38, 39, 41, 50, 52, 56, 59, 60, 62]; contributing to the interpretation and research findings [40, 59]; and subsequently dissemination to research participants [34, 38, 40, 52]; developing and testing website content [64]; interpreting study results and advising on implications of the study [38]; challenging research team's understanding and assumptions [40]; providing input into evidence-based recommendations [58], and CYP assisted with the development of training materials for doctors [53].

Impact

The NIHR defines impact as "the changes, benefits and learning gained from the insights and experiences of patients, carers and the public when working in partnership with researchers and others involved in NIHR initiatives" [66].

Based on this definition 41% of the reports (13/32) fully met the criteria describing the impact of PPI with CYP on the design of the study [34, 38–41, 53, 55–57, 59, 60, 64, 65]. Two of these reports [38, 65] provided helpful tables (see Tables 6 and 7) to map out PPI and perceived impact on the research process and intervention. 9% of the reports partially met (3/32) this criterion [49, 54, 58], and the remaining 50% (16/32) did not mention any impact.

Those that did not mention any impact focused on reporting (in varying degrees) the process of involvement and the roles CYP played within the project, without describing what difference the involvement made to the research, to CYP or to the researchers involved. For example, some reports mention CYP had helped to shape a patient information sheet but did not provide details of what happened as a result, or what changes the research team took on board. There was a lack of description on how CYP helped the design of the study, or how the young people felt about their involvement.

Reported impacts of involvement on CYP

Only 6% of the reports (2/32) had undertaken some form of evaluation to measure the impact and experiences of CYP involvement [39, 53], and one report had undertaken a narrative self-reflection of the involvement experiences over the course of ten years [56]. All three reports published findings in separate journal articles [63, 67, 68].

Despite the lack of any formal evaluations of impact, the personal impacts on CYP were reported in19% of the reports (6/32) [39, 53, 55–57, 65], with only three

including descriptions directly from CYP [39, 53, 65]. CYP felt that their involvement gave them the opportunity to meet other young people who shared the same or similar health conditions to them as one young person highlighted: "It was good to hear other people's point of view. I think it was really helpful having other people who have gone through the same things as you that understand you". They also felt that their experiences and contributions could make a real difference [65]. One young person wrote to the Principal Investigator of the study to highlight how her role had helped in her recovery from SAD [55]. One research teams reflections claimed that the benefits of being in a group gave young people the space to realise their cancer diagnosis which helped them to continue living their lives and gave young people the confidence to find their 'new normal' [56]. Reported impacts on CYP also included gaining skills such as research and technical skills [39, 56, 57], public speaking [56, 57] and confidence to interact with wider groups of young people and healthcare professionals [57, 65]. Impact on young people's vocational or employability skills were highlighted in two reports [39, 55] such as recording achievements of involvement (by receiving certificates) and helping towards applying for university. Two reports [53, 55] highlighted how CYP felt valued and listened to adding to their self-confidence.

Reported impacts on researchers

The reported impacts of involving CYP on researchers were only reported in 6% of the reports (2/32) [53, 57]. Both reports highlighted that involving CYP reinforced to research team members the importance of involving and gaining the views of CYP, ensuring that the research was of relevance to them. One report [53] simply felt that CYPs enthusiasm to be involved was motivating to team members, and one report [57] highlighted that it built their capacity to undertake effective PPI with CYP.

Dissemination

12% of the reports (4/32) fully met the criteria describing CYP involvement in the dissemination activities of the study [41, 53, 56, 65], 6% partially met (2/32) this criterion [57, 60], and the remaining 81% (26/32) did not meet this criterion. Out of those who fully met the criteria two reports [41, 53] highlighted how CYP had contributed directly to dissemination activities throughout the programme beginning at the launch of the project, during the project (writing blogs of their experiences, presentation of emerging results, being interviewed for radio) running sessions at the final dissemination event and working with a theatre group to create an artistic interpretation of emerging results, having securing additional charity funding. One report [65] highlighted how CYP had recorded materials for Podcasts (one focused on the findings from the project and one to explain their experiences of being involved). One report [41] discussed how CYP activities had helped to refine the content and framing of the dissemination activities, in particular how study findings would be communicated to members of the public. Those who partially met the criteria alluded to some form of input from CYP to assist with the dissemination of study findings, explaining for example how CYP input will support the development of plain English summaries of the results, and focused on how findings will be distributed widely (e.g., via social media).

Reflections and lessons learned

19% of the reports (6/32) fully met the criteria describing their reflections on PPI activities [41, 53, 55–57, 65], 6% partially met (2/32) this criterion [40, 60], the remaining 72% (24/32) did not meet this criterion.

Reported challenges of involving CYP

Typically, the challenges of involving CYP in PPI were linked to practical and structural issues related to the respective study, such as population issues relating to recruitment and retention issues and sustaining CYP involvement over a long period of time, which required extensive relationship building [41, 56, 57]. Accommodating CYP's availability (school commitments [60, 65], and capacity for schools to accommodate PPI requests [64] were also noted as key challenges. There was a need for more training for both PPI co-applicants and CYP [55], and recognition that explaining different methodologies to CYP can be difficult (e.g., involving young people in the design of patient information sheets was easier than explaining concepts such as core outcome set development). Structural constraints (research time frames, ethics procedures, allocated time, etc.) was noted as a key challenge [55, 56], and meeting CYP expectations particularly around the use of emerging technologies and social media which was constrained by resources and internal governance issues [56]. One report noted that if PPI was delivered by external PPI professionals independent of the research team meant that the research team were unable to discuss directly with the group which areas of feedback they were able to address, and those aspects of the study for whatever reason could not be modified [41].

Reported enablers

Typically, some reports highlighted several solutions to tackle some of the challenges, including issues around recruitment and retention. For example, some teams highlighted the need for creating wide-reaching recruitment strategies, and having pre-existing relationships and networks to access was helpful (PPI networks, charity and patient organisation support, existing groups, connections in schools) [39, 41, 53, 55-57, 64]; communicating between meetings and maintaining regular contact [53, 56, 57], and valuing the input of CYP by thanking them on a regular basis [57] was deemed beneficial. Many reports focused on the logistics of PPI with CYP, for example making meeting spaces more welcoming, flexible, and enjoyable for CYP and allowing dedicated time to spend socialising with other members [53, 55-57, 65]. This also included having a dedicated PPI budget to offer payments, reimbursement of travel expenses, funds for refreshments, and having skilled PPI leads and PPI support to deliver the PPI activities with CYP [41, 55-57]. Having access to training and inductions for CYP was also deemed beneficial [40, 41] as was having clear role descriptions [40, 55]. It was also important to develop clear project specific payment policies [40, 55-57, 65], and other relevant safeguarding policies (i.e., out of hours, sickness policies) [56], and codes of conduct (e.g., alcohol, smoking and drug use) [56]. Finally, communication that is tailored to CYP capabilities about the research itself and PPI process including feedback was viewed as important to build trust and retain their involvement [41, 57].

Discussion

This article provides an overview of reported PPI activities with CYP in 32 NIHR-funded study reports using a participation matrix, framework analysis and accompanying tools. Recording and reporting PPI is important, both to ensure transparency in relation to the contributions and roles of different stakeholders within the research process and to contribute to the evidence base within the field of PPI.

The assessment of the report's highlights that the current reporting of involvement in health research is very poor, and rarely describes who was involved (demographic details of those involved, age, sex, ethnicity, etc.), and what outcomes and impact involvement had on the research process, on CYP and/or on researchers. The analysis framework and rating tool to assess the NIHR reporting requirements identified that despite the NIHR advice and guidance for authors to follow the GRIPP2 Checklist, only a small percentage followed this. We identified significant variability in the way PPI with CYP was reported. Common themes that emerged was that those who mentioned some form of dialogue with an existing YPAG or other existing forum (i.e., school group or PPI network), as opposed to establishing their own PPI structures, tended to report less and very rarely mentioned any outcomes or impact from that involvement. Some reports provided detailed information but either blurred the lines between PPI activities with actual study methods or focused the report on adult stakeholder involvement (i.e., parent/carer involvement, healthcare professional, teacher, etc.) as opposed to CYP involvement.

To better understand the opportunities and levels of CYP involvement throughout the research process we adapted an existing participation matrix that is more commonly applied to social science research/policy and practice rather than in a health research/PPI setting. The matrix and rating tool proved to be useful to explore the reported opportunities and levels of involvement of CYP across the different phases of the research process. Whilst using these to assess reported PPI activities was difficult due to the poor level of reporting, it did allow us to make some assessment of the opportunities offered to CYP across the phases of the research process. The matrix gave us a clearer picture of the reported opportunities offered to CYP, with the majority taking place during the design phase of the research, and only a small number of reports consulting and collaborating with CYP across other phases of the research. These reports tended to be long term studies (such as programme grants funded for five years and more) with dedicated PPI personnel and consumable budget to deliver PPI across the programme. We believe that despite the difficulties using the matrix, analysis framework and rating tools on published reports, they are extremely helpful tools to plan future activities with CYP (and other populations), and can ultimately support the quality, transparency, and consistency of the reporting process that the NIHR aspires to. We believe the tools can be helpful in a number of ways including: as supportive tools to be used from inception of the research idea to inform grant applications and help identify possible involvement levels/activity prior to funding; the tools follow the research pathway so research teams can truly understand the opportunities offered to CYP and the resources required to involve them in a meaningful way; they can also be used to codevelop the PPI plan with CYP so that CYP can choose the levels of involvement across the different phases of the research process. Furthermore, the tools could be used to reflect on PPI throughout the research project, recording when and how CYP contributed, what they influenced, what changed and why this mattered to either the research project, CYP and/or researchers.

Further work is required to understand why and how the NIHR, and other funders of health research (charities, UKRI, etc.) collect such information, and what their intentions are for sharing the findings with the wider research community to continuously improve best-practice in PPI.

Strengths and Limitations

The current framework analysis had some limitations. First, limiting our search to reports submitted to the NIHR Journal Library, whilst ensuring a level of consistency, has limitations as searches in other journals or funders of different types of health research may have revealed different results. Second, the individual search terms used a large age range to identify paediatric studies which means that some reports may not have been included and this may have limited the findings. Despite these limitations, this review has strengths such as our commitment to CYP involvement, and the level of detail provided focusing on NIHR-funded studies only. The matrix, analysis framework and accompanying rating tools were deemed sufficient tools to assess the levels of involvement and opportunities offered to CYP, and assessment of how PPI with CYP described in the reports meet the NIHR reporting expectations.

Conclusion

There is a paucity of knowledge regarding involvement activities in the research process in child health research. Questions surrounding PPI are now moving beyond the 'why' to the 'how', yet the reporting of PPI (in general) leaves knowledge users with an insufficient understanding of how the work was conducted, thus limiting its reproducibility, applicability, and impact. Furthermore, shared learning from previous experiences is lost and potentially a waste of resource that could otherwise be put to ensuring 'meaningful' involvement with CYP and other patient populations.

Reporting the processes and outputs of CYP involvement more rigorously may help child health researchers involve CYP more meaningfully by learning from the experiences, enablers and challenges faced by other researchers. Furthermore, sharing information about how CYP are involved in the research cycle via different activities can help researchers in planning and conducting future studies and to reflect on their current involvement practices. Such reporting of PPI with CYP should enhance the ability to develop evidence-based guidance around how to meaningfully involve CYP in paediatric health research, and to explore and evaluate the impact of their involvement. This knowledge may also help CYP gain more awareness about the ways they can contribute as 'advisors' or 'co-researchers' and the type of influence they can have. As other publications have highlighted [21-23] more needs to be done to seek the views of CYP involved in PPI activities and the impact being involved has on them as opposed to the impact on the research only.

Supplementary Information

The online version contains supplementary material available at https://doi.org/10.1186/s40900-023-00477-8.

Additional file 1: YPAG Agenda.

Additional file 2: Summary of NIHR reports.

Additional file 3: Tasks associated to each phase of the research.

Acknowledgements

The authors are grateful healthcare professionals who are involved in the NIHR Paediatric Methodology Incubator, and to the seven young people from the GenerationR Liverpool Young Person's Advisory Group.

Author contributions

The lead author (JP) initiated the work and led the project to completion, with support from all co-authors. JP designed the workshops with healthcare professionals and GenerationR Liverpool YPAG with support from GB and VH. JP was responsible for the conception and design of the paper. JP drafted the paper and subsequent re-drafts in collaboration with GB, VH, and ED. JP prepared all tables and figures. GB redesigned Fig. 3. MB and HS critically revised the paper for important intellectual content. All authors approved the final version of the paper.

Funding

JP's time for this work was funded by the Hugh Greenwood Legacy for Children's Health Research Fund as part of a Hugh Greenwood Legacy Fellowship.

Availability of data and materials

The article is not drawing on data that is suitable to a data repository. Further information on how we facilitated the workshop meetings is available from the corresponding author.

Declarations

Ethics approval and consent to participate

We did not seek ethical approval to undertake this work.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Institute of Life Course and Medical Sciences, University of Liverpool, Liverpool, UK. ²Bristol Medical School, University of Bristol, Bristol, UK. ³Population Health Sciences Institute, University of Newcastle, Newcastle upon Tyne, UK. ⁴Rehabilitation Department, Great North Children's Hospital, Newcastle upon Tyne, UK. ⁶School of Social Justice and Law, University of Liverpool, Liverpool, UK. ⁷Department of Paediatric Rheumatology, Alder Hey Children's NHS Foundation Trust, Liverpool, UK.

Received: 8 March 2023 Accepted: 24 July 2023 Published online: 31 July 2023

References

- Department of Health. Research Governance Framework for Health and Social Care. London: Department of Health; 2005.
- 2. Department of Health, Best Research for Best Health. 2006. London.
- European Medicines Agency. Patients and consumers. 2020 [cited 2022 6th April]; Available from: https://www.ema.europa.eu/en/partnersnetworks/patients-consumers.
- 4. US Food and Drug Administration. FDA patient-focused drug development guidance series for enhancing the incorporation of the patient's voice in medical product development and regulatory decision making. 2020. [cited 2022 6 April]; Available from: https://www.fda.gov/drugs/devel opment-approval-process-drugs/fda-patient-focused-drug-developmentguidance-series-enhancing-incorporation-patients-voice-medical.
- UK Research and Innovation. Shared commitment to improve public involvement in research. 2022 [cited 2022 10th May]; Available from: https://www.ukri.org/news/shared-commitment-to-improve-publicinvolvement-in-research/.
- Patient-Centered Outcomes Research Institute (PCORI). Highlights of PCORI-Funded Research Results. 2019 [cited 2022 6 April]; Available from: https://www.pcori.org/document/highlights-pcori-funded-research-resul ts-fact-sheet.
- Clinical Trials Transformation Initiative (CTTI). CTTI Recommendations on Effective Engagement with Patient Groups Around Clinical Trials. 2015 [cited 2022 6 April]; Available from: https://ctti-clinicaltrials.org/our-work/ patient-engagement/patients-groups-clinical-trials/.
- National Standards for Public Involvement. 2019 [accessed 21 Sep 2021]; Available from: https://sites.google.com/nihr.ac.uk/pi-standards/home
- Patient Focused Medicine Development (PFMD). Patient Engagement Quality Guidance. 2020 [accessed 21 Sep 2021]; Available from: https:// synapse.pfmd.org/initiatives/the-patient-engagement-quality-guida nce-pegg.
- Tavistock and Portman NHS Foundation Trust. Patient and Public Involvement. Available: https://tavistockandportman.nhs.uk/about-us/get-invol ved/patient-and-public-involvement/ [Accessed 29 Jul 2022].
- National Institute for Health and Care Research. Briefing notes for researchers - public involvement in NHS, health and social care research. Available: https://www.nihr.ac.uk/documents/briefing-notes-for-resea rchers-public-involvement-in-nhs-health-and-social-care-research/27371 [Accessed 29 Jul 2022].
- NIHR Involve. What is public involvement in research?. Available: https:// www.invo.org.uk/find-out-more/what-is-public-involvement-in-resea rch-2/ [Accessed 29 Jul 2022].
- National Co-ordinating Centre for Public Engagement. What is public engagement? Available: https://www.publicengagement.ac.uk/aboutengagement/what-public-engagement [Accessed 29 Jul 2022].
- 14. Boyden J, Ennew J. Children in focus: a manual of participatory research with children. 1997, Save the Children.
- 15. Patton GC, et al. Our future: a Lancet commission on adolescent health and wellbeing. Lancet. 2016;387(10036):2423–78.
- UNCRC, General Comment No. 12: The right of the child to be heard. 2009.
- Brady LM, Preston J. How do we know what works? Evaluating data on the extent and impact of young people's involvement in English health research. Res All. 2020;4(2):194–206.
- Moore T, Noble-Carr D, McArthur M. Changing things for the better: the use of children and young people's reference groups in social research. Int J Soc Res Methodol. 2016;19:241–56.
- Bird D, Culley L, Lakhanpaul M. Why collaborate with children in health research: an analysis of the risks and benefits of collaboration with children. Arch Dis Childhood Educ Pract Ed. 2013;98(2):42–8.
- Jones EL, et al. Quality of reporting on patient and public involvement within surgical research: a systematic review. Ann Surg. 2015;261(2):243–50.
- 21. Wilson O, Daxenberger L, Dieudonne L, Eustace J, Hanard A, Krishnamurthi A, Quigley P, Vergou A. A rapid evidence review of young peoples involvement in health research. London: Wellcome; 2020.
- 22. Van Schelven F, et al. Patient and Public Involvement of young people with a chronic condition in projects in health and social care: a scoping review. Health Expect. 2020;23:789–801.

- Rouncefield-Swales A, et al. Children and young people's contributions to public involvement and engagement activities in health-related research: a scoping review. PLoS ONE. 2021;16(6): e0252774.
- Staniszewska S, et al. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. Res Involv Engag. 2017;3(1):13.
- Russell J, Greenhalgh T, Taylor M. Patient and public involvement in NIHR research 2006–2019: policy intentions, progress and themes. 2019.
- Moher D, et al. Guidance for developers of health research reporting guidelines. PLoS Med. 2010;7(2): e1000217.
- 27. Lansdown G, O'Kane C. A toolkit for monitoring and evaluating children's participation. 2014, Save the Children.
- Lansdown G. Conceptual framework for measuring outcomes of adolescent participation. 2018, Unicef for every child.
- Mandoh M, et al. Shifting from tokenism to meaningful adolescent participation in research for obesity prevention: a systematic scoping review. Front Public Health. 2021;9: 789535.
- Gale NK, et al. Using the framework method for the analysis of qualitative data in multi-disciplinary health research. BMC Med Res Methodol. 2013;13(1):117.
- 31. World Health Organization. Health for the world's adolescents: a second chance in the second decade: summary. 2014: Geneva.
- 32. NIHR INVOLVE. Involving children and young people as advisors in research: Top tips and essential key issues for researchers. 2019.
- 33. Blair J, et al. Continuous subcutaneous insulin infusion versus multiple daily injections in children and young people at diagnosis of type 1 diabetes: the SCIPI RCT. Health Technol Assess. 2018;22(42):1–112.
- Goodwin J, et al. Standing frames for children with cerebral palsy: a mixed-methods feasibility study. Health Technol Assess. 2018;22(50):1–232.
- Tancred T, et al. Interventions integrating health and academic education in schools to prevent substance misuse and violence: a systematic review. Public Health Res. 2019;7(17).
- Bonell C, et al. Modifying the secondary school environment to reduce bullying and aggression: the INCLUSIVE cluster RCT. Public Health Res. 2019. https://doi.org/10.3310/phr07180.
- Ramanan AV, et al. Adalimumab in combination with methotrexate for refractory uveitis associated with juvenile idiopathic arthritis: a RCT. Health Technol Assess. 2019;23(15):1–140.
- Ponsford R, et al. A school-based social-marketing intervention to promote sexual health in English secondary schools: the Positive Choices pilot cluster RCT. Public Health Res. 2021;9(1).
- Alderson H, et al. Behaviour change interventions to reduce risky substance use and improve mental health in children in care: the SOLID three-arm feasibility RCT. Public Health Res. 2020;8(13).
- Bray N, et al. Powered mobility interventions for very young children with mobility limitations to aid participation and positive development: the EMPoWER evidence synthesis. Health Technol Assess. 2020;24(50):1–194.
- Robling M, et al. The Family Nurse Partnership to reduce maltreatment and improve child health and development in young children: the BB: 2–6 routine data-linkage follow-up to earlier RCT. Public Health Res. 2021;9(2).
- Meiksin R, et al. A school intervention for 13- to 15-year-olds to prevent dating and relationship violence: the Project Respect pilot cluster RCT. Public Health Res. 2020;8:1–338.
- Mallucci CL, et al. Silver-impregnated, antibiotic-impregnated or nonimpregnated ventriculoperitoneal shunts to prevent shunt infection: the BASICS three-arm RCT. Health Technol Assess. 2020;24(17):1–114.
- 44. Caldwell DM, et al. School-based interventions to prevent anxiety, depression and conduct disorder in children and young people: a systematic review and network meta-analysis. Public Health Res. 2021;9(8).
- 45. Langton Hewer SC, et al. Intravenous or oral antibiotic treatment in adults and children with cystic fibrosis and Pseudomonas aeruginosa infection: the TORPEDO-CF RCT. Health Technol Assess. 2021;25(65):1–128.
- Kidger J, et al. Mental health support and training to improve secondary school teachers' well-being: the WISE cluster RCT. Public Health Res. 2021. https://doi.org/10.3310/phr09120.

- 47. Cottrell DJ, et al. A pragmatic randomised controlled trial and economic evaluation of family therapy versus treatment as usual for young people seen after second or subsequent episodes of self-harm: the Self-Harm Intervention - Family Therapy (SHIFT) trial. Health Technol Assess. 2018;22(12):1–222.
- Ford T et al. Training teachers in classroom management to improve mental health in primary school children: the STARS cluster RCT. Public Health Res 2019;7(6):1–150
- Janssens A, et al. The transition from children's services to adult services for young people with attention deficit hyperactivity disorder: the CATCh-uS mixed-methods study. Health Technol Assess. 2020. https://doi. org/10.3310/hsdr08420.
- Byford, S., et al., Alternative community-based models of care for young people with anorexia nervosa: the CostED national surveillance study. Health Services and Delivery Research, 2019.
- 51. Adab P, et al. The West Midlands ActiVe lifestyle and healthy Eating in School children (WAVES) study: a cluster randomised controlled trial testing the clinical effectiveness and cost-effectiveness of a multifaceted obesity prevention intervention programme targeted at children aged 6–7 years. Health Technol Assess. 2018;22(8):1–608.
- Cameron ST, et al. Provision of the progestogen-only pill by community pharmacies as bridging contraception for women receiving emergency contraception: the Bridge-it RCT. Health Technol Assess. 2021;25(27):1–92.
- Colver, A., et al., Facilitating the transition of young people with long-term conditions through health services from childhood to adulthood: the Transition research programme. Programme Grants for Applied Research, 2019.
- King C, et al. Sexual risk reduction interventions for patients attending sexual health clinics: a mixed-methods feasibility study. Health Technol Assess. 2019;23(12):1–122.
- Creswell C, et al. Cognitive therapy compared with CBT for social anxiety disorder in adolescents: a feasibility study. Health Technol Assess. 2021;25(20):1–94.
- Taylor RM, et al. Specialist cancer services for teenagers and young adults in England: BRIGHTLIGHT research programme. Programme Grants for Applied Research, 2021.
- 57. Hall NJ, et al. Conservative treatment for uncomplicated appendicitis in children: the CONTRACT feasibility study, including feasibility RCT. Health Technol Assess. 2021;25(10):1–192.
- Bee P, et al. Self-care support for children and adolescents with long-term conditions: the REFOCUS evidence synthesis. Health Services and Delivery Research, 2018.
- 59. Griffiths FE et al. The role of digital communication in patient–clinician communication for NHS providers of specialist clinical services for young people [the Long-term conditions Young people Networked Communication (LYNC) study]: a mixed-methods study. Health Services and Delivery Research. 2018.
- Giles EL, et al. Brief alcohol intervention for risky drinking in young people aged 14–15 years in secondary schools: the SIPS JR-HIGH RCT. 2019.
- Tume LN, et al. Determining optimal outcome measures in a trial investigating no routine gastric residual volume measurement in critically III children. J Parenter Enteral Nutr, 2020;n/a(n/a).
- Maguire A, et al. Best-practice prevention alone or with conventional or biological caries management for 3- to 7-year-olds: the FiCTION threearm RCT. Health Technol Assess. 2020;24(1):1–174.
- 63. Taylor RM, et al. Involving young people in BRIGHTLIGHT from study inception to secondary data analysis: insights from 10 years of user involvement. Res Involv Engag. 2018;4:50.
- Mitchell KR, et al. A peer-led intervention to promote sexual health in secondary schools: the STASH feasibility study. Public Health Res. 2020;8:1–152.
- 65. Moore DA, et al. Interventions to improve the mental health of children and young people with long-term physical conditions: linked evidence syntheses. Health Technol Assess. 2019;23(22):1–164.
- Denegri S. Definition of the impact of public involvement in health research. 2019. [Accessed 21 Sep 2022]; Available from: https://simon denegri.com/2019/07/03/downloadable.

- 67. Dovey-Pearce G, et al. The burden of proof: the process of involving young people in research. Health Expect. 2019;22(3):465–74.
- Alderson H, et al. "You've come to children that are in care and given us the opportunity to get our voices heard": the journey of looked after children and researchers in developing a Patient and Public Involvement group. Health Expect. 2019;22(4):657–65.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

